Dear editor and Reviewers:

Thank you for your letter and the reviewers’ comments concerning our manuscript entitled “Microarray analysis to explore the effect of CXCL12 isoforms in a pancreatic pre-tumor cell model” (Manuscript number: 69240). These comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. We have revised and resubmit the revised manuscript. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

Reviewer

Well written manuscript. Accept

Author response and action taken:

Thank you very much for your careful and meticulous work.

EDITORIAL OFFICE’S COMMENTS

3. Academic norms and rules: The signed Conflict-of-Interest Disclosure Form and Copyright License Agreement are not provided but are necessary.

Author response and action taken: We have provided The signed Conflict-of-Interest Disclosure Form and Copyright License Agreement.
We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Yours sincerely,
Yan-Dong Miao
miaoyd19@lzu.edu.cn